MOSS 1 Kick-off Meeting Agenda

April 1, 2003

MOSS meeting to be held today at 2pm in bldg 32 room S241.

For those dialing in the information follows.

DIAL-IN NUMBER:

  USA Toll Free Number: 888-395-8106

  PASSCODE: 92854

Agenda:

1) MOSS 1 dates:

MOSS 1 to start April 28; Data duration up to 24 hours but varies by type

MOSS 1 dry run proposed for on or after April 15th

2) Aura Science Testing Website: http://jupiter02.gsfc.nasa.gov/auratesting/

a) Action Item tool

b) Science Data Flow diagrams

c) Contact List

d) Schedule

e) Requirements

f) MOSS TRR and Debrief templates will be provided prior to TRR and Debrief

Element Status:

EDOS

-Action Item tool POC's: Beth Nolletti (password needed) (add Ron?)

-Processing of MOSS1 data MLS, HIRDLS and TES  completed

-MOSS 1 requests for EDOS processing: GBAD data, OMI data (can SCIF2 data be supported with April 15th data?)

-Plans for SCIF 3 data distribution  (VC's 6,7,8,9 not captured to tape,GBAD?); MLS engineering data in VC8 used for science data processing

EMOS/FOT

-Action Item tool POC's:  Ernest Canevari  (password needed) (add Utah & also FOT POC?)

-GBAD data generated for MOSS 1: Completed; Data transfer to Eric Chomko pending then conversion to CADU and tape prepared for EDOS

-Generation of Carry-out files and transfer to GDAAC using MOSS 1 data as well as SCIF 3 data; set a date for testing this interface (conduct ICT at the same time)

FDS

-Action Item tool POC's: Elizabeth Corderman (password needed) and Lauri Newman (password needed)

-Products provided for GBAD generation: completed; orbital elements were corrected by FDS; Correct orbital elements to be provided for use with Orbsim

-Definitive Ephemeris delivery to GDAAC for MOSS 1  April 15th noon-noon

-Use of real interface not planned in MOSS 1; data to be provided and staged

LDAAC

-Action Item tool POC's: Steve Protack (any other?)

-OPS mode confirmed for MOSS 1; SIPS product transfers to DAAC OPS?

-SCIF 3 data receipt readiness

-ESDT's --any issues?

-Verify subscriptions are active in OPS mode

TES

-Action Item POC's: Rob Toaz(password needed) and Andy Griffin(password needed)

-Data: Ready; Processed by EDOS; Delivered to SIPS and processed

-MOSS 1 processing plans

-Reminder to save April 15th Auxiliary data

-Any test support needed prior to the Dry run?

GDAAC

-Action Item tool POC's: Ed McDonald (any other?)

-Run DPREP upon receipt of FDS orbit and EMOS GBAD carryout ; distribute DPREP'd data to SIPS

-SCIF 3 data receipt readiness

-ESDT's --any issues?

-Verify subscriptions are active

MLS

-Action Item tool POC's: David Cuddy and Audrey Hanzel

-Data: Time shifted and Ready; Processed by EDOS; Delivered to SIPS and processed

-MOSS 1 processing plans

-Reminder to save April 15th Auxiliary data

-Any test support needed prior to the Dry run?

HIRDLS

-Action Item tool POC's: Ken Stone (password needed) and Vince Dean

-Data: Time shifted and Ready; Processed by EDOS; Delivered to SIPS and processed

-MOSS 1 processing plans

-Reminder to save April 15th Auxiliary data

-Any test support needed prior to the Dry run?

OMI SIPS

-Action Item tool POC's: Marghi Hopkins and Ted Johnson

-Working on product transfers to GDAAC for ICT6

-MOSS 1 data; some SCIF 2 PDS data obtained from GDAAC; Carol is working on timeshifting routine for 2 timecodes and checksum

-Plan to use SCIF data in MOSS 1 without timeshifting due to schedule constraints

-Eric to obtain data from GDAAC or directly from EDOS tape?

-MOSS 1 processing plans; June processing test

ODPS: Separate Telecon to be held due to time difference with Netherlands

-Action Item tool POC's: Bert van den Oord and Jon van de Vegte

-ODPS have established MTMGW accounts

Averstar

-Action Item tool POC's: Dot Cutcher and Scott Applebee

MRTT

-Action Item tool use?

-MOSS 2 preparation

-Data for GN testing
